

Temi di Discussione

(Working Papers)

Political selection in the skilled city

by Antonio Accetturo





Temi di discussione

(Working papers)

Political selection in the skilled city

by Antonio Accetturo

Number 956 - March 2014

The purpose of the Temi di discussione series is to promote the circulation of working papers prepared within the Bank of Italy or presented in Bank seminars by outside economists with the aim of stimulating comments and suggestions.

The views expressed in the articles are those of the authors and do not involve the responsibility of the Bank.

Editorial Board: Giuseppe Ferrero, Pietro Tommasino, Margherita Bottero, Giuseppe Cappelletti, Francesco D'Amuri, Stefano Federico, Alessandro Notarpietro, Roberto Piazza, Concetta Rondinelli, Martino Tasso, Giordano Zevi. *Editorial Assistants:* Roberto Marano, Nicoletta Olivanti.

ISSN 1594-7939 (print) ISSN 2281-3950 (online)

Printed by the Printing and Publishing Division of the Bank of Italy

POLITICAL SELECTION IN THE SKILLED CITY

by Antonio Accetturo*

Abstract

This paper studies the impact of citizens' human capital on the characteristics of elected politicians in democratic elections for the post of mayor. By using a change in the rules for Italian mayoral elections and a difference-in-differences estimator, I find that cities endowed with a larger amount of human capital tend to elect mayors that have a higher educational attainment and that were previously employed in skill-intensive jobs. This result is quantitatively small but robust to omitted variables or selection issues.

JEL Classification: D7, I20, H8.

Keywords: human capital externalities, political selection, cities.

Contents

1.	Introduction	5
2.	How do (did) Italians elect their mayors?	8
	2.1 New rules	9
	2.2 Overall assessment	10
3.	Empirical design	12
4.	Data and variables	14
5.	Results	15
	5.1 Baseline results	16
	5.2 Robustness	17
	5.3 An interpretational issue: selection or homophily?	23
6.	Concluding remarks	24
Re	eferences	26
Та	Tables and figures	

^{*} Bank of Italy, Economic Research Unit, Trento Branch.

1. Introduction¹

This paper deals with the benefits of schooling. It aims to establish whether better-educated citizens make different political choices with respect to persons with a lower amount of human capital.²

The fact that a pool of better-educated individuals may exert a positive effect on political outcomes is well-known in the field of Political Science. As Almond and Verba (1989) point out, education is a crucial determinant of "civic culture" and, as regards participation in democratic politics, "the uneducated man [...] is a different political actor from the man who achieved a higher level of education".

The linkage between schooling and political choices rests on the idea that education raises the cognitive skills of individuals and lowers the cost to them of acquiring information about policies and politicians. This may have two main consequences:

- 1. Political participation (turnout at elections) should be higher among educated individuals.
- 2. Political choices (in terms of preferred candidates) by educated citizens might differ from those by individuals with a lower amount of human capital.

A few papers analyse the first consequence (Dee, 2004; Milligan et al., 2004; Jaitman, 2013). The aim of this paper is to provide the first causal investigation of the second consequence. This is done by studying the political selection of Italian mayors over the

¹ I wish to thank Mino Barone, Matteo Bugamelli, Gustavo Canavire-Bacarreza, Giuseppe Ferrero, Francesco Manaresi, Sauro Mocetti, Massimo Morelli, Pierluigi Murro, Elisabetta Olivieri, Wouter Vermeulen, two anonymous reviewers, participants at the following conferences/seminars: AIEL (Rome, September 2013), SIE (Bologna, October 2013), and IRVAPP (Trento, November 2013) for their useful suggestions, and Jennifer Parkinson for editorial assistance. The usual disclaimers apply. Further, the views expressed are the author's own and do not necessarily reflect those of the Bank of Italy.

² In this paper, I use education and human capital interchangeably. Education is one of the most popular topics of analysis for economists and studies on its impact on individual wages and productivity date back more than four decades. However, its positive effects may spill over the individual sphere: economists have recently argued the existence of positive externalities due to the mere presence of a more educated pool of workers on aggregate productivity. These are defined as social returns to education or human capital externalities. Moretti (2004) extensively reviews this literature.

period 1987-1997 and assessing, in particular, whether municipalities endowed with higher human capital end up electing politicians with higher education or a more skill intensive job.

I concentrate on the second consequence for two main reasons. The first relates to the importance of political leadership. Electing talented individuals is of primary importance for the functioning of democratic systems and for the successful adoption of important economic reforms. As Besley (2005) points out "[in the past 200 years,] Almost every major episode of economic change [...] has been associated with key personalities coming to power with a commitment to these changes."³ The second reason relates to the relative strengths of the two consequences under different institutional settings. As Milligan et al. (2004) suggest, the effects of education on turnout are limited to the cases in which voting is particularly costly (i.e. pre-registration for the US). This implies that it is quite unlikely to find an effect of schooling on participation in areas in which costs are negligible as is the case, for example, of many European countries (where local administrations are in charge of voter registration). This may mean that, in these countries, the second consequence is comparatively stronger than the first one and, for this reason, it deserves a thorough empirical investigation.

The identification of a causal link between citizens' education to political selection is not an easy task. Human capital is not randomly distributed across cities: it tends to be higher in places with high productivity, good amenities, and better institutions. This implies that better administered cities are likely to attract a disproportionate number of more educated citizens, thus creating a severe problem of reverse causality (Glaeser and Gottlieb, 2009). Moreover, local politicians are usually selected from a pool of local candidatecitizens (Besley and Coate, 1997) thus implying a spurious correlation between the two variables.

In order to identify a causal parameter, I make use of a difference-in-differences (diffin-diffs) framework that exploits a change in the electoral rules for mayors in 1993. In the pre-1993 context, mayors were elected by city councils; local elections were dominated by

 $^{^{3}}$ Key (1956) remarks that "The nature of the workings of government depends ultimately on the men who run it. The men we elect to office and the circumstances we create that affect their work determine the nature of popular government. Let there be emphasis on those we elect to office."

city-level sections of national parties and voters had little command over the lists that these parties presented. As a result, voters' decisions were mainly *ideological* and their attention was directed more towards parties' political platforms rather than the characteristics of the candidates themselves. The 1993 reform established direct voting for mayors along with new powers. As a consequence, voters were more likely to evaluate the characteristics of the candidates more carefully, alongside making the usual assessment of their political platforms. Politicians' characteristics are measured by years of schooling, which signals for an individual's intrinsic abilities, and skill intensity of her previous profession that proxies "market skills" which can reasonably be correlated with political abilities (Caselli and Morelli, 2004). The core of my identification strategy rests on the fact that if better-educated individuals prefer candidates with a larger amount of human capital, the rise in the educational attainment of mayors should be stronger in cities that are more endowed in terms of human capital. The use of first-differencing helps to control for all other time-invariant characteristics and has clear methodological advantages with respect to the previous literature that relies, instead, on cross-sectional identifications.⁴

Results show that average education of voters positively impacts on the schooling and the skill intensity of previous profession of elected mayors. This result is confirmed in a number of robustness checks that take into account omitted variables bias or sorting of citizens. The estimated effect does not seem to be driven by homophily. The magnitude seems instead comparatively small: the interquantile range of the human capital distribution across cities accounts for 6% of the interquantile rise in the schooling of mayors due to the 1993 reform mentioned above.

This is the first paper that tries to uncover the causal link between citizens' educational level and political selection. Previous studies mostly concentrated on the relationship between schooling and the stability of democratic systems from both a theoretical (Bourguignon and Verdier, 2000; Glaeser et al., 2007) and an empirical point of view (Barro, 1999; Glaeser et al., 2004; Papaioannou and Siourounis, 2008). A much closer stream of research investigates, instead, the relationship between turnout at elections and

⁴ This kind of identification is widely used for the analysis of the heterogeneous effects of trade liberalizations on firm-level outcomes (see, for example, Verhoogen, 2008).

education. Dee (2004) finds that civic awareness is positively influenced by schooling. The result holds by using college proximity and exposure to child labour laws as instruments for school attendance. Milligan et al. (2004) use an instrumental variable strategy based on the changes in compulsory schooling laws and find a strong effect of education on voting in presidential elections over the period 1948-2000. They also find that educated adults are more likely to discuss politics and associate with political groups. The effect, however, is limited to the cases in which registration for voting lies with the citizen. When it rests on local government officials (as in the UK), they fail to find any effect. Jaitman (2013) analyses the relationship between turnout and education by exploiting, in a regression discontinuity framework, a feature of the Argentinean electoral law according to which voting is compulsory (and enforced by fines) for individuals aged below 70.⁵

This paper also relates to the "Political Selection" literature, that analyses the determinants of the attraction and retention of talented individuals into politics. To the best of my knowledge, there are no works linking citizens' education to the selection of politicians.⁶

The paper is organized as follows. Section 2 introduces the institutional framework. Section 3 describes the empirical design. Section 4 presents the data and some descriptive statistics. Section 5 shows the results. Section 6 concludes.

2. How do (did) Italians elect their mayors?

Local governance in Italy is a 3-tier system: 20 regions, 110 provinces, and roughly 8,100 municipalities. Municipal administrations are in charge, among other things, of managing public utilities (local roads, water, and garbage collection), the provision of public elementary schooling, and care of children and the elderly. This is why Italian citizens are

⁵ A somewhat related stream of literature has focused on the 1993 reform in Italy and its political effects. A widely exploited feature is that cities with more than 15,000 inhabitants have run-offs while for smaller cities a first-past-the-post mechanism is at work (see Section 2). Bordignon et al. (2011) show the positive effects of the dual ballot rule on the number of political candidates. De Paola and Scoppa (2010) analyse, instead, the effects of the number of candidates on the quality of elected politicians. Barone and De Blasio (2013) show the positive effects of the dual ballot on turnout; they also find that a dual ballot rule improves the quality of the politicians selected for the executive body.

⁶ In the robustness check section (5.2.2), where I list a set of possible confounding factors that are likely to influence my estimations, I will briefly review the "Political Selection" literature.

generally very interested in the composition and the performance of municipal administrations. These comprise an executive body, made up of the mayor and the executive committee, and a legislative body (the municipal council).

2.1 New rules

A major breakthrough in city-level governance was achieved by the 1993 electoral reform: new rules radically changed the political arena for the election of mayors, representing a drastic policy shift.⁷

Table 1 gives an overview of the main changes involved:

Mayors. - Before 1993, mayors were elected by municipal Councils, generally from among the available councilors. As a standard practice in Italian politics, political majorities in the municipal council were created after the elections; this implied that the choice of a mayor was based on bargaining among the local political party leaders following the results of the poll. According to a long-standing tradition, mayors were not the local party leaders.⁸ After 1993, direct elections were established. Different rules applied according to the size of the city measured in terms of number of inhabitants. For municipalities with more than 15,000 inhabitants, there was a run-off between the two candidates with the most votes if no single candidate received 50 per cent of the ballot. For smaller cities a first-past-the-post system was in place. As a standard practice, in municipalities with fewer than 5,000 inhabitants candidates ran their own lists (called Liste Civiche) without any formal affiliation to political parties. This was due to the fact that in those cities, even before the reform, a majority premium was granted to the most voted party, thus generating an incentive to aggregate parties into coalition lists. The 1993 reform extended the mayors' powers: members of the executive committee were now directly appointed by the mayor, who could even recruit local experts without political affiliations.

⁷ Law 81/1993, *Elezione diretta del sindaco e del presidente della provincia* (Direct election of mayors and provincial presidents). Previously, elections had been regulated by Presidential Decree (DPR) 570/1960 and Law 663/1964.

⁸ The system was actually mirroring that of central government, established by the 1948 Constitution, with the explicit aim of preventing the creation of a strong executive power - a clear reaction to the 1922-1943 fascist regime.

The municipal council. – Before 1993, the municipal council was at centre stage in local politics. It was the only directly elected body and could appoint or remove the mayor. All elections were held with a proportional system, except for cities with fewer than 5,000 inhabitants in which majoritarian elections were held. The 1993 reform marked a sharp reduction in the powers of the municipal councils as it could no longer appoint the mayor and the executive committee. Moreover, a motion of no confidence in the mayor automatically implied the dissolution of the council. Lastly, the 1993 reform extended the majoritarian system to all cities, irrespective of size.⁹

After the approval of the reform in March 1993, the new rules began to be implemented gradually according to the schedule for new elections envisaged at local level.

2.2 Overall assessment

Pre-1993 politics was characterized by the overwhelming influence of national political parties. It was nearly impossible to identify a possible mayor before the elections, since postelection bargaining was particularly delicate and no party could seriously commit to choosing a possible candidate. This was an intended outcome: the system was aimed at ensuring close party control over local governmental activities and – at the same time – at preventing the mayor from consolidating any degree of autonomous leadership. Strom (1990) reports that Italian politics was, at that time, characterized by a low level of "ex-ante identifiability" as it was nearly impossible, before elections, to predict the winning candidate for the head of the executive. As a result, the intermediation by political parties was overwhelming and voting was predominantly ideological as little room was left for the local politicians to improve their visibility.¹⁰

⁹ It is not easy to understand the kind of political selection that was working in the pre-reform system. The paper by Mattozzi and Merlo (2011), however, is able to give some guidance on how parties select their members and possible candidates for elections. They find that parties have weak incentives to select individuals with very high abilities since a relatively more homogeneous group (without superstars) is likely to maximize the collective effort of the party. In other words, selection by parties leads to a set of candidates whose characteristics are much closer to the median voter's characteristics.

¹⁰ For similar scenarios, at local level, see also Agosta (1999), Parisi (1984), Pasquino (2001), and Vandelli (1997).

The 1993 reform was passed in a period of great transformation in Italian politics. In 1990, following years of political paralysis, massive government debt, extensive corruption, and the considerable influence of organized crime, there was a great demand for political, economic, and ethical changes. The reform was introduced to reinvigorate local governments and to simplify local administration. The target was to meet a broader demand for accountability and limit the overwhelming influence of political parties.

As for its stated goal, the reform is deemed a success; it managed to introduce a new (and more stable) system of representation. Voters had more influence and more incentives to carefully select their mayors. According to Baldini and Legnante (1998), the new system secured a more visible and independent position for the new mayors: "by being directly elected above and beyond the party wrangling in the municipal council and bureaucratic fetters, the mayor gains individual legitimation and visibility which are valuable instruments not so much for governing, but particularly for re-election."

A first, non causal, assessment of the effects of the reform is given in Figure 1. For each election year after 1993, I report the share of winning mayors running with their own party list without any formal affiliation to national parties (blue bar). For each year, I also report the same figure for their (pre-1993) predecessors (red bar).¹¹ It is apparent that the 1993 reform markedly reduced ideological voting, as mayors were now directly relying on their own platforms and personal characteristics.

This is reflected by a rise in the educational attainment of elected mayors: in Figure 2, I show the average increase in years of schooling of the elected mayor with respect to their predecessor for three cases. For the first case, the blue bars represent the increase for individuals elected under the pre-1993 electoral system; for the second case, the red bars show the increase from a pre-reform to a post-reform system; and for the third case, the green bars represent the rise in education of mayors elected under the post-1993 regime. The change in schooling for mayors elected under the same regime (both before and after, blue and green bars) does not display a clear increasing pattern, thus suggesting that, for those

¹¹ Independent mayors are those that either are labelled as "Independent" in the Register of Local Administrators or were elected heading a *Lista Civica*. The sample includes cities larger than 5,000 since, all Mayors in smaller cities usually ran independently.

elections, incentives to select politicians remained stable. A stark difference emerges, instead, when we compare mayors elected under different electoral rules (red bars): in this case, incentives to choose a more educated politician rose and this is reflected by a steady increase in the average years of schooling of mayors.

3. Empirical design

A typical way to identify a causal effect in the human capital externalities literature is to use some institutional factors that determine (exogenous) differences in local human capital accumulation as an instrument for the current endowment of city-average schooling (Milligan et al., 2004; Acemoglu and Angrist, 1999). In this paper, I make use of an alternative strategy based on a diff-in-diffs framework.

More formally the baseline diff-in-diffs model is as follows:

$$y_{ct} = \alpha_c + \delta_1 TIME_t + \delta_2 School_c * POST_t + \delta_3 X_{ct} + \xi_{ct}$$
(1)

where y_{ct} is either average schooling or a dummy equal to one for individuals with a skill intensive job for mayors elected in city *c* at time *t*. α_c is a city fixed effect, *TIME*_t is a set dummy variables equal to one for all election years, *POST*_t is a dummy equal to one for all elections that took place after 1993, *School*_c is the treatment variable and indicates the educational attainment (human capital) at city level, and X_{ct} is a set of time-varying city-level controls.

As Bertrand et al. (2004) point out, standard errors of equation (1) may be systematically downward biased if serial autocorrelation is not seriously taken into account.¹² Bertrand et al. (2004) propose a number of solutions. The simplest one, which I use in this paper, is to collapse the information on two periods, before and after the reform, by taking averages of all variables and running a regression over two points in time.

¹² Note that the use of the city-level fixed effect does not solve the problem since fixed effects do not take into account the autocorrelation of the $School_c * POST_t$ term.

Averages are calculated over the period 1987-92 for the pre-reform years and 1993-97 for the post-reform period.¹³

Equation (1) changes as follows:

$$\overline{y}_{ct} = \alpha_c + \beta_1 POST_t + \beta_2 School_c * POST_t + \beta_3 \overline{X}_{ct} + \varepsilon_{ct} \qquad t = 0,1$$
(2)

where \bar{y}_{ct} and \bar{X}_{ct} are the average of, respectively, the dependent variable and controls before and after the reform.

After taking the averages, I compute the first-differences of equation (2) and estimate this simple equation:

$$\Delta \overline{y}_c = \beta_1 + \beta_2 School_c + \beta_3 \Delta \overline{X}_{ct} + \varepsilon_{ct}$$
(3)

Where $\Delta \overline{y}_c$ and $\Delta \overline{X}_{ct}$ are the changes over time of the averaged dependent and explanatory variables. The coefficient of interest is β_2 that aims to capture the effect of human capital on the rise in schooling (of in the skill intensity of the previous job) of elected mayors.

There are three main conditions under which β_2 can be deemed as a causal parameter:

- The treatment variable (*School_c*) is predetermined with respect to the reform (predetermined treatment).
- There are no other omitted variables that correlate with both the schooling level of the population and the change in the characteristics of elected mayors after 1993 (omitted variable bias).
- Changes in the dependent variable are not correlated with *School_c* before the implementation of the reform (parallel trend assumption).

I deal with these issues in the robustness check section.

¹³ Five years is the maximum term of a local legislature.

4. Data and Variables

The data come from two main sources: the Register of Local Administrators (RLA) and the 1991 and 1971 population censuses.

The RLA is managed by the Central Directorate of Electoral Services of the Italian Ministry of the Interior. The register provides is a picture of all elected officers in all Italian municipalities, provinces, and regions at 31 December of each year starting from 1985. For municipalities, "elected officers" include the mayor, deputy mayor, members of the Executive Committee, the chair and the vice-chair of the city council, and all councilors. For cities in which no mayor is currently in office, ¹⁴ the database also indicates the Commissioner appointed by the Ministry of the Interior before new elections. The database is quite detailed. It indicates year of election, party political affiliation, date and city of birth, educational attainment, and previous occupation.

The RLA provides the dependent variable of equation (3).

As for the characteristics of the mayor, I use two measures:

- *Education.* This proxies for individual intrinsic abilities. It is computed by taking the logs of the minimum years required to obtain the individual's highest degree. As usual in the estimates of Mincerian equations for Italy, I attribute: zero years for no formal education (uncompleted elementary school);
 5 years for elementary school; 8 years for lower secondary (middle) school; 13 years for upper secondary (high) school; 18 years for a university degree. Alternatively I use the share of elected Mayors with at least a secondary school (high) diploma.
- *Skill intensity* of the Mayor's previous occupation. It proxies for "market skills," which, according to the literature dealing with political selection, can be highly correlated with political abilities (Caselli and Morelli, 2004). It is computed as the share of individuals with a high-skilled job before and after

¹⁴ This can occur for several reasons: resignation, death, forced dissolution of the city council due to suspected organized crime infiltration.

1993. Skilled workers are: university professors, journalists, physicians, pharmacists, judges, lawyers, engineers, chemists, biologists, architects, mathematicians, economists, statisticians, business consultants, and managers.¹⁵

The other main data sources are the Italian censuses of 1991 and 1971. They provide for the treatment variable, that is the educational level of the resident population (over the age of 6) at city level. 1991 data is used in the baseline regression, 1971 data is used as an instrument in a robustness check. In this case too, I calculate the average schooling of the population by using the lowest number of years necessary to obtain the higher degree.

Table 2 reports the descriptive statistics of all the variables used in the empirical analysis.

As Table 2 shows, schooling of mayors rose by 6.4%, the share of individuals previously working in skill-intensive professions increased by 2.7 percentage points, the share of high-school graduates grew by 6.6 percentage points after the reform. The average number of years of schooling of mayors before 1993 (13.3, not reported) only just corresponds to the achievement of a high school diploma and is more than twice the average number of years of schooling of the population in 1991 (6.5).

5. Results

In this section, I show the results by assessing whether human capital externalities have an effect on the political selection at city level.

The core of my identification strategy is the diff-in-diffs estimation presented in Section 3. In Section 5.1 I show the baseline results on the average schooling of elected mayors. In Section 5.2 I make a number of robustness checks that take into account issues like omitted variable bias, sorting, and the parallel trend assumption. Further robustness is achieved by considering alternative time spans and making a placebo experiment. Finally, I show that the main result holds when using alternative measures.

¹⁵ For similar definitions see Brollo et al. (2013) and De Paola and Scoppa (2010).

5.1 Baseline results

Estimates of equation (3) are displayed in Table 3. Column 1 reports the results for the full sample that includes all Italian cities. The coefficient of interest (β_2) is positive and significant with a point estimate of 0.077. This means that, for a municipality at the 75th percentile of the human capital distribution, the 1993 reform raised the schooling of elected mayors by 7 per cent while the effect for a municipality at the 25th percentile was 5.8 per cent. This means that the implied effect of the interquatile range (7-5.8=1.2 percentage points) accounts for roughly 6% of the interquantile range of the dependent variable (0.207). The implied effect is quite small, which is not surprising as the period under consideration was characterized by radical changes in Italian politics that may have affected other determinants of political selection. However, as we will see, the result is very stable and robust, thus confirming the existence of human capital externalities in the selection of politicians.¹⁶

As shown in Table 1, the 1993 reform envisaged different electoral mechanisms according to city size. We take into these issues into account in columns (2) and (3) by looking at two population thresholds, one regarding single/dual ballots and the other concerning party political affiliations.

The first threshold is set at 15,000. According to the new rules, in municipalities smaller than this, all elections are with a first-past-the post system, and the winner is the one who wins more votes. For larger cities, electoral rules envisage a second round if none of the candidates received 50% of the votes in the first round. As pointed out by De Paola and Scoppa (2010) and Barone and de Blasio (2013), this is likely to change the nature of political competition: a dual ballot, indeed, raises the overall schooling of candidates by triggering broader political representation and more competition among political parties.

¹⁶ The negative (but not significant) sign for the constant should not be misinterpreted. It indicates the potential effect of the reform for a municipality with the log of the average schooling of population equal to zero (i.e. with an average schooling of population of one year). By calculating the implied effect for the city at the 5th percentile of the human capital distribution (i.e. with the log of schooling of population equal to 1.63, 5.13 years), the implied effect is positive and equal to 4.8 per cent.

The second threshold is set at 5,000. As a standard practice even before the reform, candidates in cities smaller that size were not usually affiliated to national parties but ran their own party list (called *liste civiche*). This was due to the fact that electoral rules envisaged a majority premium for the most voted party and groups belonging to the same coalition had strong incentives to gather in the same list. This is a relevant feature under direct elections as it is likely to reduce the potential influence of national politics and raise the importance of individual characteristics.

Estimates are provided in columns (2) and (3) for, respectively, the 15,000 and 5,000 inhabitants threshold. The baseline result is confirmed: estimated value of β_2 is positive and significant and even slightly larger than the baseline one.¹⁷

5.2 Robustness

5.2.1 Predetermined treatment

The causal nature of the estimates of β_2 heavily relies on the predetermined nature of the treatment variable with respect to the policy change (see Section 3). In this study, the treatment variable (level of schooling of the population) is measured in 1991, two years before the variation of the electoral rules (1993). This should be enough as long as the 1993 reform could not be predicted in 1991 and the population did not start sorting according to education before 1993.

In order to reject this possibility, I instrument the 1991 average schooling of the population at its 1971 level. This is a good instrument assuming that the 1993 reform could not be predicted in 1971, which is a more than reasonable assumption.

The results are displayed in Table 4. First stage statistics are quite satisfactory: the F of excluded instruments is very high and well above the rule-of-thumb value of 10. The coefficient of the first stage is also positive and highly significant, thus indicating a good correlation between current and past values of educational levels at city level. The second stage results are also quite reassuring. The coefficient is positive, significant, and quite

¹⁷ The implied effects for both thresholds are quite similar to the baseline one as the interquantile ranges for the restricted samples are slightly smaller than for the full sample.

similar in magnitude with respect to the OLS values, with the single exception of the estimate for cities with fewer than 5,000 inhabitants, that is slightly larger.

5.2.2 Omitted variable bias

Causality also rests on the absence of omitted variables that are correlated with both the educational level of the population and the growth in the schooling of elected mayors. Political selection literature identifies a number of determinants of the political selection. Table 5 presents a brief review of the literature and also reports the variables that I use as controls.

A first concern relates the Besley and Coate (1997) candidate-citizen model, according to which candidates self-select from the pool of citizens.¹⁸ This implies a possible spurious correlation between the education of the citizens and the characteristics of candidates. This feature is controlled by first-differencing and by using a predetermined treatment variable - see eq. (3).

The second issue relates to changes in economic conditions. Caselli and Morelli (2004) show that, when returns to education are particularly high in the private sector, politicians with a high human capital may decide to stay out of politics thus leaving room for less educated candidates.¹⁹ Empirically, my main concern relates to the possible changes in returns to education in the period under consideration due, for example, to skill-biased technical change (Acemoglu, 1998; Autor et al., 2003). This feature may create an attenuation bias in the estimate of β_2 if those technologies were more likely to be adopted in cities with higher human capital (Accetturo et al., 2013). In order to capture this issue, I use

¹⁸ See also Osborne and Slivinski (1996).

¹⁹ This effect is amplified when the observable characteristic is particularly noisy in signaling the true quality of an individual. Dal Bò et al. (2006) extended the Caselli-Morelli analysis to give a role to pressure groups.

the 1991 and 1996 economic censuses to control for changes in the labour demand for more educated workers: Δln (*employment – skilled*).²⁰

The third possible confounder is related to the arrival of public funds and their distortionary effects in the selection of politicians (Brollo et al., 2013). Larger transfers may increase the probability for mismanagement and rent extraction, thus raising the relative returns for being in office for lower ability individuals. Eventually, this may tilt the political equilibrium by inducing a sustained entry for less educated candidates. From an empirical point of view, if the eligibility for transfers were stable over time, first-differencing would cope with this problem. However, this is not the case for Italy: the early 1990s were characterized by a thorough review of the public interventions in lagging areas (Braunerhjelm et al., 2000; Accetturo and de Blasio, 2012). In particular, in the 1980s and before 1992, aid to the underdeveloped South was based on direct transfers to households and basically all municipalities received them. After 1992, aid was reorganized according to the regulations envisaged by Law 488/1992 (Bronzini and De Blasio, 2006). The new rules established an explicit targeting of productive units (mainly industrial plants). This had two consequences: (i) only a subset of southern cities succeeded in keeping their transfers; (ii) for the first time some cities in the de-industrializing North were able to obtain public funds. order to control for this change, I insert as an explanatory variable In $\Delta Transfers_i = D_{i, pre-1992} - D_{i, post-1992}$, where $D_{i, pre-1992}$ and $D_{i, post-1992}$ are a dummy variable equal to one if city *i* received transfers, respectively, before and after 1992.²¹

The fourth issue relates to the change in the political conditions at city level. Galasso and Nannicini (2011) show that higher contestability generates better political selection.²² As explained in Section 2, the 1993 electoral reform was introduced in a period of increasing contestability within the Italian political system, due, mainly, to both left- and right-wing new political parties having access to the political arena. If the increase in contestability

²⁰ Highly skilled industries are selected by using the EU-KLEMS database and are defined as the industries that, over the period 1991-96, employed a higher share of individuals with a university degree than the Italian average. These industries are: Education (Nace rev. 1: M), Health (N), Financial intermediation (J), Renting and business activities (K), and Other public administration (L).

²¹ Data on transfers under Law 488/92 were kindly provided by the Ministry of Economic Development.

²² This is confirmed in a subsequent paper (Galasso and Nannicini, 2012) for different electoral rules.

were equally spread across all municipalities, its effect would simply be absorbed by the constant. If, instead, it were more pronounced in more educated cities, estimates would be upwardly biased. In order to tackle this issue, I insert a measure of political contestability taken from general election results ($\Delta Contestability$).²³ I construct a variable that is the difference in the share of votes between the winning and the second party at city level in the general elections of 1987 and 1992 (for the years before the change) and 1994 and 1996 (for the years after). This variable is subsequently first-differenced. For all the available years, I took the votes in the proportional elections.²⁴

Finally, I also insert a dummy variable equal to one for the Special Status Regions (SSRs), i.e. the five autonomous regions (Sicily, Sardinia, Valle d'Aosta, Trentino-Alto Adige, and Friuli-Venezia Giulia) that established slightly different rules for local elections.²⁵

Results are displayed in Table 6 and confirm the baseline estimates of Table 3. The measure of contestability is never statistically significant, thus suggesting that changes in contestability equally affected all municipalities. Moreover the controls for high-skilled sectors and transfers are not significant.

Are these controls enough to reject an omitted variable bias? In order to answer this question, I assess the relative importance of unobservable omitted variables by analysing the possible variation in the coefficient of interest with the inclusion of observable explanatory variables. If these controls substantially attenuate the estimates, it is possible that inclusion of more controls would drive the estimated effect to zero. Conversely, if the inclusion of

²³ Data are taken from the Istituto Carlo Cattaneo (ICC) database on national Italian elections from 1861 to 2008. ICC is a private research organization aimed at promoting studies on political issues in Italy. ICC runs a huge database on the results of political elections at municipal level.

²⁴ This is a relevant issue for the 1994 and 1996 general elections. For those years, a mixed electoral system was at work. Some of the members of Parliament were elected in majoritarian constituencies, while the remaining part of Parliament was elected under a proportional system. All voters were given two voting papers, one for the majoritarian vote and the other for the proportional vote. In the calculation of this measure, I only take the votes for the proportional seats. All elections before 1994, instead, were conducted under a purely proportional system.

²⁵ Local rules mostly involved the way in which the municipal councils were elected and the appointment rules for the Executive Committee (see Baldini and Legnante, 1998).

controls does not change the point estimate in a significant way, I can more confidently claim the causal interpretation of the parameter.

Following Altonji et al. (2005) and its extension to the continuous case made by Bellows and Miguel (2009), I measure the relative strength that omitted variables should have vis-à-vis the observed controls to completely drive the coefficient to zero. This is calculated as the ratio between the coefficient of interest with controls (for example, 0.071 for the full sample case) and the coefficient without controls (0.079, this is the coefficient on the same estimation sample, as reported at the foot of the table) and the coefficient with controls (0.071). The results for these calculations are reported in the last row of the table. If the set of observed controls is representative of all possible controls, then a large ratio suggests that it is implausible that omitted variable bias can explain away the entire effect. In the full sample case, unobserved omitted variables should be 8.88 times stronger than observed controls to drive the coefficient to zero, which seems highly unlikely.²⁶

5.2.3 Parallel trend assumption

A third threat to the causal interpretation of β_2 rests on the fact that an increase in the schooling of mayors may have started even before the 1993 reform for more educated cities, thus violating the parallel trend assumption.

In order to address this issue I split the sample in two. The first subperiod relates to the pre-treatment period (1987-1992), the second to the years 1993-1997. For both subperiods, I run eq. (1) by using each point in time as a (placebo) treatment year. Due to the correlation between city fixed effects and $School_c * POST_t$, regressions cannot be run for the years 1987 and 1993 for, respectively, the 1987-1992 and 1993-1997 periods.

Figure 3 reports the coefficient and the 95% confidence intervals for δ_2 . For comparative reasons I also report the 1993 estimate, calculated over the entire period 1987-1997. If δ_2 were positive and significant before or after 1993, the parallel trend assumption

²⁶ Altonji et al. (2005) consider 3.55 a fairly safe ratio for their estimates.

would be violated thus invalidating my identification strategy. Results show that this is not the case as the only significant coefficient is that related to 1993.

5.2.4 Alternative time span and (placebo) effects on councilors

I also further explore the robustness of the results by changing the time span of analysis. In Table 7, I restrict the analysis to the period 1990-96; I use the 1990-92 years as pre-reform and 1993-96 as post-reform. Sample size is now reduced as some cities did not elect a mayor both before and after 1993 in this restricted period. The baseline results still hold: the estimated coefficients are now slightly larger but confidence intervals largely overlap with the baseline result.

Table 8 presents a placebo experiment on the average schooling of councilors. As explained in Section 3,²⁷ the 1993 reform affected the way mayors were elected but they dealt only marginally with city councils. In particular, rules did not change at all for cities with fewer than 5,000 inhabitants (Table 1) while, for larger municipalities, elections passed from a proportional to a majoritarian system. The coefficient of interest is the one in column (3) (i.e. for councils unaffected by the reform) and is small and far from standard significance levels. This shows that political selection did not change significantly more in cities with a high level of human capital with respect to the others. This result is also confirmed for larger cities - see columns (1) and (2).

5.2.5 Alternative measures

In Table 9, I report the baseline estimates for other measures commonly used in the empirical literature on political selection: the share of mayors with a high-skilled job and the share of mayors with at least a secondary education. Results show that even by using the alternative definition, the baseline result is confirmed.

²⁷ See also Bordignon et al. (2011).

5.3 An interpretational issue: selection or homophily?

Results so far undoubtedly point to the fact that cities endowed with more human capital end up in electing more educated politicians. A possible concern in the interpretation of the results rests with the use of schooling of mayor as dependent variable. Educated citizens may not prefer more educated politicians because their expected abilities as administrators, but just because they are closer in terms of preferences and characteristics. This means that the causal effect uncovered by this paper is just linked to homophily rather than selection.

If homophily is at work, I should observe a reversion to population mean of elected mayors after the reform. In other words, once allowed to directly choose their preferred candidate, citizens should tend to elect politicians that are closer (in terms of educational attainment) to population average.

A simple way to cope with this issue is to assess the existence of heterogeneous effects across cities. In principle, under homophily, cities with a lower endowment of human capital should observe a decrease in the average schooling of elected mayors; this is due to the fact that, given the fuzzy nature of political selection before the reform, mayors in those municipalities were likely to be much more educated than local population. In more skilled cities, instead, where mayors were likely to have an educational level comparable with the citizens' one, the increase should be close to zero.²⁸

In order to detect heterogeneity I discretise the schooling level of population into four quartiles and run equation (3).

Results, displayed in Table 10, report, for each quartile, both the coefficient and a measure of political mismatch for each group.²⁹

²⁸ Alternatively, in highly educated cities, elected mayors before 1993 could be less educated than population average. In this case, the increase in this group should be positive and significant.

²⁹ Political mismatch is computed as $100 * (School_{mayor, before, c} / School_{c} - 1)$, that is the distance (in percentage points) between the years of schooling of mayors elected before 1993 and the educational attainment of the population.

Let us first focus on the first quartile. Before 1993 this groups was electing mayors with a schooling level 158%-159% higher than local population (see the mismatch indicator). Under homophily, the growth in schooling for that group should be negative, as citizens should prefer a mayor that is closer in terms of educational attainment. This is not confirmed by data: the dummy for the first quartile is positive and significant in all specification, thus implying that even low-skilled/heavily mismatched cities did not observe a deterioration in the schooling of elected politicians after the reform.

The analysis of the fourth quartile points to the same direction. Cities in that group were comparatively less mismatched than others as mayors elected before 1993 had, on average, only 72-81 per cent of schooling more than population. Under homophily, this implies that those citizens had fewer incentives to change the educational level of the elected mayor and the coefficient for that quartile should be lower than those of the other groups. This is not supported by results, as the coefficient for the fourth quartile is larger in a statistically significant way than those of the other groups.³⁰

6. Concluding remarks

The fact that more educated citizens behave differently from uneducated individuals in terms of political choices has sustained a long-lasting debate in the field of Political Science and Political Economy. Empirical evidence on this, however, is quite scant. This paper has filled this gap by analysing a new area over which education may exert its effects: the selection of politicians.

In this study I found that cities with a greater endowment of human capital tend to elect mayors with a higher educational attainment and that were previously employed in skill-intensive jobs. For this purpose, I exploited the 1993 electoral reform of the system of electing Italian mayors that generated stronger incentives to evaluate the mayor's characteristics in a diff-in-diffs framework. The estimated effect is quantitatively small, but robust to omitted variable and citizen sorting biases; it does not seem to be driven by

³⁰ These results are confirmed by a local polynomial estimations (available upon request).

homophily. Overall, this finding has far-reaching consequences for economic policies and, in particular, for the educational policies by governments.

References

- Accetturo A. and G. de Blasio (2012): "Policies for local development: an evaluation of Italy's "Patti Territoriali,"" *Regional Science and Urban Economics*, vol. 42, pp. 15-26.
- Accetturo A., A. Dalmazzo, and G. de Blasio (2013): "Skill Polarization in Local Labour Markets under Share-Altering Technical Change," forthcoming *Journal of Regional Science*.
- Acemoglu, D. (1998): "Why Do New Technologies Complement Skills? Directed Technical Change And Wage Inequality," *The Quarterly Journal of Economics*, vol. 113, pp. 1055-1089.
- Acemoglu, D. and J. Angrist (1999): "How Large are the Social Returns to Education? Evidence from Compulsory Schooling Laws," NBER Working Papers 7444, National Bureau of Economic Research, Inc.
- Agosta, A. (1999): "Sistema elettorale e governo locale: Gli effetti politici e istituzionali della riforma del 1993," in S. Operto (ed.) Votare in città. Riflessioni sulle elezioni amministrative, Milan, Franco Angeli, pp. 31–58.
- Almond, G. and S. Verba (1989): "The Civic Culture: Political Attitudes and Democracy in Five Nations," Sage Publications.
- Altonji J.G., Elder T., Taber C. (2005): "Selection on observed and unobserved variables: assessing the effectiveness of catholic schools" *Journal of Political Economy*, 113, pp. 151–184.
- Autor D., F. Levy, and R. Murnane (2003): "The skill content of recent technological change: an empirical exploration," *The Quarterly Journal of Economics*, vol. 118, pp. 1279-1333.
- Baldini G. and G. Legnante (1998): "The 1997 Local Elections in Italy: From party Mayors to a Mayors' Party," in L. Bardi and M. Rhodes (eds): Italian Politics. Oxford: Westview Press.
- Barone G. and G. de Blasio (2013): "Electoral rules and voter turnout," *International Review* of Law and Economics, vol. 36, pp. 25-35.
- Barro, R. (1999): "Determinants of democracy," *Journal of Political Economy*, vol. 107, pp. 158–183.
- Bellows J. and E. Miguel (2009): "War and local collective action in Sierra Leone," *Journal* of Public Economics, vol. 93, n. 11-12, pp. 1144-1157.
- Bertrand, M., E. Duflo, E., and S. Mullainathan (2004): "How Much Should We Trust Differences-in-Differences Estimates?," *The Quarterly Journal of Economics*, vol. 119, pp. 249-275.
- Besley, T. (2005): "Political Selection", *Journal of Economic Perspectives*, vol. 19, n. 3, pp. 43-60.

- Besley, T. and S. Coate (1997): "An Economic Model of Representative Democracy," *The Quarterly Journal of Economics*, vol. 112(1), pp. 85-114.
- Bordignon, M., T. Nannicini, and G. Tabellini (2011): "Moderating political extremism: Single round vs. runoff elections under plurality rule" mimeo.
- Bourguignon, F., and T. Verdier (2000): "Oligarchy, democracy, inequality, and growth," *Journal of Development Economics*, vol. 62, pp. 285–313.
- Braunerhjelm, P., R. Faini, V. Norman, F. Ruane, and P. Seabright (2000): "Integration and the Regions of Europe: How the Right Policies Can Prevent Polarization," Centre for Economic Policy Research, London.
- Brollo, F., T. Nannicini, R. Perotti and G. Tabellini (2013): "The political resource curse," Forthcoming, *American Economic Review*.
- Bronzini, R. and G. de Blasio (2006): "Evaluating the impact of investment incentives: The case of Italy's law 488/1992," *Journal of Urban Economics*, vol. 60, pp. 327–349.
- Caselli, F. and M. Morelli (2004): "Bad Politicians," *Journal of Public Economics*, vol. 88(3-4), pp. 759-82.
- Dal Bo, E., P. Dal Bo, and R. Di Tella (2006): "Plata o Plomo? Bribe and Punishment in a Theory of Political Influence," *American Political Science Review*, vol. 100(1), pp. 41–53.
- De Paola, M. and V. Scoppa (2010): "Political competition and politician quality: Evidence from Italian municipalities," *Public Choice*, vol. 148, pp. 547–559.
- Dee, T. (2004): "Are there civic returns to education?," *Journal of Public Economics*, vol. 88, pp. 1696–1712.
- Galasso V. and T. Nannicini (2011): "Competing on Good Politicians," *American Political Science Review*, vol. 105, pp. 79-99.
- Galasso V. and T. Nannicini (2012): "Political Selection under Alternative Electoral Rules," mimeo.
- Glaeser, E., R. LaPorta, F. Lopez-de-Silanes, and A. Shleifer (2004): "Do institutions cause growth?", *Journal of Economic Growth*, vol. 9, pp. 271–303.
- Glaeser, E. and J. Gottlieb (2009): "The Wealth of Cities: Agglomeration Economies and Spatial Equilibrium in the United States," *Journal of Economic Literature*, vol. 47(4), pp. 983-1028.
- Glaeser, E., G. Ponzetto, and A. Shleifer (2007): "Why does democracy need education?," *Journal of Economic Growth*, vol. 12(2), pp. 77-99.
- Jaitman L. (2013) "The causal effect of compulsory voting laws on turnout: Does skill matter?", *Journal of Economic Behavior and Organization*, vol. 92, pp. 79-93.
- Key, V. (1956): "American State Politics: An Introduction," New York: Alfred A. Knopf.

- Mattozzi A. and A. Merlo (2011): "Mediocracy, Fourth Version," PIER Working Paper Archive 13-010, Penn Institute for Economic Research, Department of Economics, University of Pennsylvania, revised 08 Feb 2013.
- Milligan, K., E. Moretti, and P. Oreopoulos (2004): "Does education improve citizenship? Evidence from the United States and the United Kingdom," *Journal of Public Economics*, vol. 88(9-10), pp. 1667-1695.
- Moretti, E. (2004): "Human capital externalities in cities," Handbook of Regional and Urban Economics, in: J. V. Henderson and J. F. Thisse (ed.), Handbook of Regional and Urban Economics, edition 1, volume 4, chapter 51, pp. 2243-2291 Elsevier.
- Osborne, M. and A. Slivinski (1996): "A Model of Political Competition with Citizen Candidates," *The Quarterly Journal of Economics*, vol. 111, pp. 65–96.
- Papaioannou, E. and G. Siourounis (2008): "Democratisation and Growth," *Economic Journal*, vol. 118(532), pp. 1520-1551.
- Parisi L. (1984): "Luoghi e misure della politica," Bologna, Il Mulino.
- Pasquino G. (2001): "Le elezioni comunali nella transizione italiana: La costruzione della leadership," *Teoria Politica*, vol. 17(2), pp. 91–106;
- Strom K. (1990): "Minority Government and Majority Rule," Cambridge University Press.
- Vandelli L. (1997): "Sindaci e miti. Sisifo, Tantalo e Damocle nell'amministrazione locale," Bologna, Il Mulino.
- Verhoogen E. (2008): "Trade, Quality Upgrading, and Wage Inequality in the Mexican Manufacturing Sector," *The Quarterly Journal of Economics*, vol. 123, n. 2, pp. 489-530.



SHARE OF INDEPENDENT MAYORS IN CITIES WITH >5,000 INHABITANTS

Source: Author's calculations based on data in the *Anagrafe Amministratori Locali* (Register of local administrators).

Blue bars indicate the share of "Independent" mayors in each year. The red bars show the same share for his/her (pre-1993) predecessor. "Independent" mayors are those whose main supporting party is a *Lista Civica* (a civic list). The sample only considers cities with more than 5,000 inhabitants.



SCHOOLING OF MAYORS AND ELECTORAL REFORM

Growth rate of schooling within the post-1993 electoral system

Source: Author's calculations based on data in the Anagrafe Amministratori Locali (Register of local administrators).

Growth rate of the schooling of the elected mayor in each election year with respect to his/her predecessor. Blue bars indicate the change in schooling for elections in the pre-1993 system. Red bars indicate the change in schooling from a pre-1993 to a post-1993 mayor. Green bars indicate the growth rate in schooling for elections in the post-1993 system.



COMMON TREND ASSUMPTION

Source: Author's calculations based on data in the Anagrafe Amministratori Locali (Register of local administrators) and on the 1991 census. OLS estimates, see eq. (1). Dependent variable: log of the mayor's schooling. Robust 95% c.i. reported. For the years 1988-92 and 1994-97, all regressions are run, respectively, over the period 1987-92 and 1993-1997. For these estimates, each point in time represents a (placebo) treatment year. 1987 and 1993 estimates are omitted due to the perfect collinearity between the $School_c * POST_t$ term and city-level fixed effects. The 1993 estimate is run over the period 1987-1997 and is reported for comparative reasons. All regressions include city and year-of-election fixed effects.

Table 1

	Before 1993	After 1993				
Mayor:						
pop<5,000	Elected by the council among council members	Direct Election (first-past-the-post)				
5,000 <pop<15,000< td=""><td>Elected by the council among council members</td><td>Direct Election (first-past-the-post)</td></pop<15,000<>	Elected by the council among council members	Direct Election (first-past-the-post)				
pop>15,000	Elected by the council among council members	Direct Election (run-off)				
Municipal Council:						
pop<5,000	Majoritarian System	Majoritarian System				
5,000 <pop<15,000< td=""><td>Proportional System</td><td>Majoritarian System</td></pop<15,000<>	Proportional System	Majoritarian System				
pop>15,000	Proportional System	Majoritarian System				
Source: DPR 570/60 and Law 663/64 for the pre-1993 system; Law 81/93 for the post-1993 system.						

THE 1993 REFORM

DESCRIPTIVE STATISTICS

	No. Obs.	Mean	Standard Deviation
$\Delta \overline{y}_c$: ln(schooling)	7,332	0.064	0.324
$\Delta \overline{y}_c$: profession	7,332	0.027	0.505
$\Delta \overline{y}_c$: high school	7,332	0.066	0.375
In(Schooling) – population (1971)	7,332	1.366	0.228
In(Schooling) – population (1991)	7,332	1.857	0.123
∆ln (employment – skilled)	7,332	0.000	0.545
∆Contestability	7,212	-0.089	0.149
∆Transfers	7,332	0.005	0.532
$\Delta \overline{y}_c$: ln(schooling) (Municipal Council)	7,291	0.068	0.146

Source: Author's calculations based on data in the *Anagrafe Amministratori Locali* (Register of local administrators), the 1971, 1991, and 1996 censuses, and ICC.

Dependent variable: In(schooling) – Mayor	Full sample	Pop<15,000	Pop<5,000				
β ₂	0.077**	0.086**	0.085**				
	(0.029)	(0.032)	(0.037)				
Constant	-0.078	-0.095	-0.094				
	(0.054)	(0.059)	(0.002)				
R^2	0.000	0.001	0.001				
No. Obs.	7,332	6,741	5,312				
Courses Authorite colouintie	بام مبم امم مما ممي	te in the America	fo America introductori				

BASELINE RESULTS

Source: Author's calculations based on data in the Anagrafe Amministratori

Locali (Register of local administrators), and the 1991 census. OLS estimates, see eq. (3). Dependent variable: growth rate of the mayor's schooling. Robust standard errors in parenthesis. * significant at 10%, ** significant at 5%, *** significant at 1%.
Dependent variable: In (schooling) – Mayor	Full sample	Pop<15,000	Pop<5,000
β ₂	0.073*	0.088**	0.125**
	(0.039)	(0.044)	(0.051)
Constant	-0.071	-0.098	-0.169*
	(0.073)	(0.080)	(0.093)
R^2	0.038	0.036	0.032
No. Obs.	7,332	6,741	5,312
F-first stage	9,144	7,904	5,759
Marginal effect (instrument)	0.412***	0.397***	0.385***
	(0.004)	(0.004)	(0.005)
Source: Author's calculations based on data in the Anagrafe Amministratori			

PREDETERMINED TREATMENT: IV ESTIMATES

Source: Author's calculations based on data in the *Anagrafe Amministratori Locali* (Register of local administrators), and the 1991 and 1971 censuses. IV estimates, see eq. (3). Dependent variable: growth rate of the mayor's schooling. Robust standard errors in parenthesis. Log schooling of population in 1991 instrumented with log schooling of population in 1971. * significant at 10%, ** significant at 5%, *** significant at 1%.

ALTERNATIVE EXPLANATIONS: WHAT DOES THE POLITICAL SELECTION LITERATURE SAY

Paper:	Economic/Political mechanism	Control variables in the regression	
Besley-Coate (1997)	Candidates self-select from the citizens pool	First Differencing and predetermined treatment variable	
Caselli-Morelli (2004); Dal Bò et al. (2006)	Economic conditions (for talented individuals) influence the entry for good politicians	∆ln (employment – skilled)	
Brollo et al. (2013)	Transfers induce negative political selection (due to rent extraction)	∆Transfers	
Galasso-Nannicini (2011 and 2012); De Paola and Scoppa (2010)	Higher contestability increases political selection	∆Contestability	
Source: Author's own literature review.			

CONTROL FOR CONTESTABILITY, TRANSFERS, ECONOMIC CONDITIONS, AND SPECIAL STATUS REGION

Dependent variable: In(schooling) – Mayor	Full sample	Pop<15,000	Pop<5,000
β ₂ (A)	0.071**	0.077**	0.081**
	(0.032)	(0.035)	(0.041)
∆ln (employment –	0.007	0.008	0.007
skilled)	(0.008)	(0.008)	(0.008)
∆Contestability	0.033	0.031	0.017
	(0.026)	(0.027)	(0.030)
Δ Transfers	-0.003	-0.002	-0.002
	(0.007)	(0.008)	(0.009)
SSR	0.000	0.000	0.006
	(0.011)	(0.012)	(0.014)
Constant	-0.065	-0.077	-0.089
	(0.060)	(0.065)	(0.076)
R^2	0.001	0.002	0.001
No. Obs.	7,212	6,624	5,200
Result of baseline estimates on the same estimation sample β_2 (B) 0.079^{**} 0.089^{**} 0.089^{**}			

β ₂ (B)	0.079**	0.089**	0.089**
	(0.029)	(0.032)	(0.037)
(A)/[(B)-(A)]	8.88	6.42	11.57
Courses Authorite	a a la vilatia na la a a a la a	data in the American	Americaintentent

Source: Author's calculations based on data in the *Anagrafe Amministratori Locali* (Register of local administrators), the 1991 and 1996 censuses, and ICC dataset.

OLS estimates, see eq. (3). Dependent variable: growth rate of the mayor's schooling. Robust standard errors in parenthesis. * significant at 10%, ** significant at 5%, *** significant at 1%.

Dependent variable: In(schooling) – Mayor	Full sample	Pop<15,000	Pop<5,000	
β ₂	0.094**	0.101**	0.097**	
	(0.032)	(0.036)	(0.041)	
Constant	-0.121**	-0.136**	-0.130**	
	(0.060)	(0.066)	(0.076)	
R^2	0.001	0.001	0.001	
No. Obs.	6,438	5,895	4,618	
Courses Authorite colouintie	بام مبم امم ممما ممي	المستحمل مطلحه أسلح	Co. A use use in is two to ui	

TIME SPAN: 1990-96

Source: Author's calculations based on data in the Anagrafe Amministratori

Locali (Register of local administrators), and the 1991 census. OLS estimates, see eq. (3). Dependent variable: growth rate of the mayor's schooling. Robust standard errors in parenthesis. * significant at 10%, ** significant at 5%, *** significant at 1%.

Table 8

Dependent variable: In(schooling)–Councilors	Full sample	Pop<15,000	Pop<5,000	
β ₂	-0.038**	0.000	0.022	
	(0.012)	(0.015)	(0.017)	
Constant	0.141***	0.078**	0.048	
	(0.026)	(0.028)	(0.032)	
R^2	0.003	0.002	0.001	
No. Obs.	7,291	6,700	5,272	
Source: Author's calculation	ons based on da	ta in the Anagra	fe Amministratori	

PLACEBO: EFFECTS ON COUCILLORS

Source: Author's calculations based on data in the *Anagrafe Amministratori Locali* (Register of local administrators), and the 1991 census.

OLS estimates, see eq. (3). Dependent variable: growth rate of the schooling of councilors. All regressions include a dummy for Special Status Regions. Robust standard errors in parenthesis. * significant at 10%, ** significant at 5%, *** significant at 1%.

Table 9

ALTERNATIVE MEASURES

	Full sample	Pop<15,000	Pop<5,000
	Share of high-sk	illed jobs	
β ₂	0.158**	0.144**	0.118**
	(0.048)	(0.052)	(0.059)
Constant	-0.266**	-0.243**	-0.200*
	(0.090)	(0.096)	(0.108)
R^2 – within	0.001	0.001	0.001
No. Obs.	7,332	6,741	5,313
	Share of secondary	education	
β ₂	0.091**	0.108**	0.112**
	(0.032)	(0.036)	(0.042)
Constant	-0.104*	-0.135**	-0.138*
	(0.061)	(0.067)	(0.077)
R^2	0.001	0.001	0.001
No. Obs.	7,332	6,741	5,313
Source: Author's calculations based on data in the <i>Anagrafe Amministratori</i> <i>Locali</i> (Register of local administrators), and the 1991 census.			

OLS estimates, see eq. (3). Robust standard errors in parenthesis. ' significant at 10%, ** significant at 5%, *** significant at 1%.

Table 10

Dependent variable: In(schooling) – Mayor	Full sample	Pop<15,000	Pop<5,000
Constant (1 st quartile)	0.050***	0.048***	0.044***
	(0.007)	(0.007)	(0.008)
Dummy 2 nd quartile	0.018*	0.018	0.022*
	(0.010)	(0.011)	(0.012)
Dummy 3 rd quartile	0.013	0.015	0.020
	(0.011)	(0.011)	(0.013)
Dummy 4 th quartile	0.026**	0.031**	0.032**
	(0.010)	(0.011)	(0.014)
Mismatch (1 st quartile)	159%	159%	158%
Mismatch (2 nd quartile)	110%	109%	104%
Mismatch (3 rd quartile)	91%	90%	85%
Mismatch (4 th quartile)	81%	77%	72%
R^2	0.001	0.001	0.001
No. Obs.	7,332	6,741	5,313
Source: Author's calculations based on data in the <i>Anagrafe Amministratori Locali</i> (Register of local administrators), and the 1991 census. Mismatch indicator is calculated as $100 * (School_{mavor,before,c} / School_c - 1)$. OLS			

SELECTION OR HOMOPHILY?

estimates, see eq. (3). Robust standard errors in parenthesis. * significant at 10%, ** significant at 5%, *** significant at 1%.

- N. 930 Uncertainty and heterogeneity in factor models forecasting, by Matteo Luciani and Libero Monteforte (September 2013).
- N. 931 *Economic insecurity and fertility intentions: the case of Italy*, by Francesca Modena, Concetta Rondinelli and Fabio Sabatini (September 2013).
- N. 932 The role of regulation on entry: evidence from the Italian provinces, by Francesco Bripi (September 2013).
- N. 933 *The management of interest rate risk during the crisis: evidence from Italian banks*, by Lucia Esposito, Andrea Nobili and Tiziano Ropele (September 2013).
- N. 934 *Central bank and government in a speculative attack model*, by Giuseppe Cappelletti and Lucia Esposito (September 2013).
- N. 935 *Ita-coin: a new coincident indicator for the Italian economy*, by Valentina Aprigliano and Lorenzo Bencivelli (October 2013).
- N. 936 *The Italian financial cycle: 1861-2011*, by Riccardo De Bonis and Andrea Silvestrini (October 2013).
- N. 937 The effect of tax enforcement on tax morale, by Antonio Filippin, Carlo V. Fiorio and Eliana Viviano (October 2013).
- N. 938 *Tax deferral and mutual fund inflows: evidence from a quasi-natural experiment*, by Giuseppe Cappelletti, Giovanni Guazzarotti and Pietro Tommasino (November 2013).
- N. 939 Shadow banks and macroeconomic instability, by Roland Meeks, Benjamin Nelson and Piergiorgio Alessandri (November 2013).
- N. 940 Heterogeneous firms and credit frictions: a general equilibrium analysis of market entry decisions, by Sara Formai (November 2013).
- N. 941 The trend-cycle decomposition of output and the Phillips curve: Bayesian estimates for Italy, by Fabio Busetti and Michele Caivano (November 2013).
- N. 942 Supply tightening or lack of demand? An analysis of credit developments during the Lehman Brothers and the sovereign debt crises, by Paolo Del Giovane, Andrea Nobili and Federico Maria Signoretti (November 2013).
- N. 943 Sovereign risk, monetary policy and fiscal multipliers: a structural model-based assessment, by Alberto Locarno, Alessandro Notarpietro and Massimiliano Pisani (November 2013).
- N. 944 *Calibrating the Italian smile with time-varying volatility and heavy-tailed models*, by Michele Leonardo Bianchi, Frank J. Fabozzi and Svetlozar T. Rachev (January 2014).
- N. 945 Simple banking: profitability and the yield curve, by Piergiorgio Alessandri and Benjamin Nelson (January 2014).
- N. 946 Information acquisition and learning from prices over the business cycle, by Taneli Mäkinen and Björn Ohl (January 2014).
- N. 947 *Time series models with an EGB2 conditional distribution*, by Michele Caivano and Andrew Harvey (January 2014).
- N. 948 Trade and finance: is there more than just 'trade finance'? Evidence from matched bank-firm data, by Silvia Del Prete and Stefano Federico (January 2014).
- N. 949 *Natural disasters, growth and institutions: a tale of two earthquakes*, by Guglielmo Barone and Sauro Mocetti (January 2014).
- N. 950 *The cost of firms' debt financing and the global financial crisis*, by Daniele Pianeselli and Andrea Zaghini (February 2014).
- N. 951 On bank credit risk: systemic or bank-specific? Evidence from the US and UK, by Junye Li and Gabriele Zinna (February 2014).
- N. 952 School cheating and social capital, by Marco Paccagnella and Paolo Sestito (February 2014).

^(*) Requests for copies should be sent to:

Banca d'Italia – Servizio Studi di struttura economica e finanziaria – Divisione Biblioteca e Archivio storico – Via Nazionale, 91 – 00184 Rome – (fax 0039 06 47922059). They are available on the Internet www.bancaditalia.it.

- A. PRATI and M. SBRACIA, Uncertainty and currency crises: evidence from survey data, Journal of Monetary Economics, v, 57, 6, pp. 668-681, **TD No. 446 (July 2002).**
- L. MONTEFORTE and S. SIVIERO, *The Economic Consequences of Euro Area Modelling Shortcuts*, Applied Economics, v. 42, 19-21, pp. 2399-2415, **TD No. 458 (December 2002).**
- S. MAGRI, *Debt maturity choice of nonpublic Italian firms*, Journal of Money, Credit, and Banking, v.42, 2-3, pp. 443-463, **TD No. 574 (January 2006).**
- G. DE BLASIO and G. NUZZO, *Historical traditions of civicness and local economic development*, Journal of Regional Science, v. 50, 4, pp. 833-857, **TD No. 591 (May 2006).**
- E. IOSSA and G. PALUMBO, *Over-optimism and lender liability in the consumer credit market*, Oxford Economic Papers, v. 62, 2, pp. 374-394, **TD No. 598 (September 2006).**
- S. NERI and A. NOBILI, *The transmission of US monetary policy to the euro area,* International Finance, v. 13, 1, pp. 55-78, **TD No. 606 (December 2006).**
- F. ALTISSIMO, R. CRISTADORO, M. FORNI, M. LIPPI and G. VERONESE, *New Eurocoin: Tracking Economic Growth in Real Time*, Review of Economics and Statistics, v. 92, 4, pp. 1024-1034, **TD No. 631 (June 2007).**
- U. ALBERTAZZI and L. GAMBACORTA, *Bank profitability and taxation*, Journal of Banking and Finance, v. 34, 11, pp. 2801-2810, **TD No. 649** (November 2007).
- L. GAMBACORTA and C. ROSSI, *Modelling bank lending in the euro area: a nonlinear approach*, Applied Financial Economics, v. 20, 14, pp. 1099-1112 ,**TD No. 650** (November 2007).
- M. IACOVIELLO and S. NERI, *Housing market spillovers: evidence from an estimated DSGE model,* American Economic Journal: Macroeconomics, v. 2, 2, pp. 125-164, **TD No. 659 (January 2008).**
- F. BALASSONE, F. MAURA and S. ZOTTERI, *Cyclical asymmetry in fiscal variables in the EU*, Empirica, **TD** No. 671, v. 37, 4, pp. 381-402 (June 2008).
- F. D'AMURI, GIANMARCO I.P. OTTAVIANO and G. PERI, *The labor market impact of immigration on the western german labor market in the 1990s*, European Economic Review, v. 54, 4, pp. 550-570, **TD No. 687 (August 2008).**
- A. ACCETTURO, Agglomeration and growth: the effects of commuting costs, Papers in Regional Science, v. 89, 1, pp. 173-190, **TD No. 688 (September 2008).**
- S. NOBILI and G. PALAZZO, *Explaining and forecasting bond risk premiums*, Financial Analysts Journal, v. 66, 4, pp. 67-82, **TD No. 689 (September 2008).**
- A. B. ATKINSON and A. BRANDOLINI, *On analysing the world distribution of income*, World Bank Economic Review, v. 24, 1, pp. 1-37, **TD No. 701 (January 2009).**
- R. CAPPARIELLO and R. ZIZZA, Dropping the Books and Working Off the Books, Labour, v. 24, 2, pp. 139-162, **TD No. 702 (January 2009).**
- C. NICOLETTI and C. RONDINELLI, *The (mis)specification of discrete duration models with unobserved heterogeneity: a Monte Carlo study*, Journal of Econometrics, v. 159, 1, pp. 1-13, **TD No. 705** (March 2009).
- L. FORNI, A. GERALI and M. PISANI, *Macroeconomic effects of greater competition in the service sector: the case of Italy*, Macroeconomic Dynamics, v. 14, 5, pp. 677-708, **TD No. 706** (March 2009).
- Y. ALTUNBAS, L. GAMBACORTA and D. MARQUÉS-IBÁÑEZ, *Bank risk and monetary policy*, Journal of Financial Stability, v. 6, 3, pp. 121-129, **TD No. 712** (May 2009).
- V. DI GIACINTO, G. MICUCCI and P. MONTANARO, Dynamic macroeconomic effects of public capital: evidence from regional Italian data, Giornale degli economisti e annali di economia, v. 69, 1, pp. 29-66, TD No. 733 (November 2009).
- F. COLUMBA, L. GAMBACORTA and P. E. MISTRULLI, *Mutual Guarantee institutions and small business finance*, Journal of Financial Stability, v. 6, 1, pp. 45-54, **TD No. 735** (November 2009).
- A. GERALI, S. NERI, L. SESSA and F. M. SIGNORETTI, *Credit and banking in a DSGE model of the Euro Area,* Journal of Money, Credit and Banking, v. 42, 6, pp. 107-141, **TD No. 740** (January 2010).
- M. AFFINITO and E. TAGLIAFERRI, *Why do (or did?) banks securitize their loans? Evidence from Italy*, Journal of Financial Stability, v. 6, 4, pp. 189-202, **TD No. 741 (January 2010).**
- S. FEDERICO, Outsourcing versus integration at home or abroad and firm heterogeneity, Empirica, v. 37, 1, pp. 47-63, **TD No. 742** (February 2010).

- V. DI GIACINTO, *On vector autoregressive modeling in space and time*, Journal of Geographical Systems, v. 12, 2, pp. 125-154, **TD No. 746 (February 2010).**
- L. FORNI, A. GERALI and M. PISANI, *The macroeconomics of fiscal consolidations in euro area countries,* Journal of Economic Dynamics and Control, v. 34, 9, pp. 1791-1812, **TD No. 747** (March 2010).
- S. MOCETTI and C. PORELLO, *How does immigration affect native internal mobility? new evidence from Italy*, Regional Science and Urban Economics, v. 40, 6, pp. 427-439, **TD No. 748 (March 2010)**.
- A. DI CESARE and G. GUAZZAROTTI, An analysis of the determinants of credit default swap spread changes before and during the subprime financial turmoil, Journal of Current Issues in Finance, Business and Economics, v. 3, 4, pp., **TD No. 749** (March 2010).
- P. CIPOLLONE, P. MONTANARO and P. SESTITO, Value-added measures in Italian high schools: problems and findings, Giornale degli economisti e annali di economia, v. 69, 2, pp. 81-114, TD No. 754 (March 2010).
- A. BRANDOLINI, S. MAGRI and T. M SMEEDING, Asset-based measurement of poverty, Journal of Policy Analysis and Management, v. 29, 2, pp. 267-284, **TD No. 755** (March 2010).
- G. CAPPELLETTI, A Note on rationalizability and restrictions on beliefs, The B.E. Journal of Theoretical Economics, v. 10, 1, pp. 1-11, **TD No. 757** (April 2010).
- S. DI ADDARIO and D. VURI, Entrepreneurship and market size. the case of young college graduates in *Italy*, Labour Economics, v. 17, 5, pp. 848-858, **TD No. 775 (September 2010).**
- A. CALZA and A. ZAGHINI, *Sectoral money demand and the great disinflation in the US*, Journal of Money, Credit, and Banking, v. 42, 8, pp. 1663-1678, **TD No. 785 (January 2011).**

2011

- S. DI ADDARIO, *Job search in thick markets*, Journal of Urban Economics, v. 69, 3, pp. 303-318, **TD No.** 605 (December 2006).
- F. SCHIVARDI and E. VIVIANO, *Entry barriers in retail trade*, Economic Journal, v. 121, 551, pp. 145-170, **TD** No. 616 (February 2007).
- G. FERRERO, A. NOBILI and P. PASSIGLIA, Assessing excess liquidity in the Euro Area: the role of sectoral distribution of money, Applied Economics, v. 43, 23, pp. 3213-3230, **TD No. 627** (April 2007).
- P. E. MISTRULLI, Assessing financial contagion in the interbank market: maximum entropy versus observed interbank lending patterns, Journal of Banking & Finance, v. 35, 5, pp. 1114-1127, TD No. 641 (September 2007).
- E. CIAPANNA, *Directed matching with endogenous markov probability: clients or competitors?*, The RAND Journal of Economics, v. 42, 1, pp. 92-120, **TD No. 665 (April 2008).**
- M. BUGAMELLI and F. PATERNÒ, *Output growth volatility and remittances*, Economica, v. 78, 311, pp. 480-500, **TD No. 673 (June 2008).**
- V. DI GIACINTO e M. PAGNINI, Local and global agglomeration patterns: two econometrics-based indicators, Regional Science and Urban Economics, v. 41, 3, pp. 266-280, **TD No. 674 (June 2008)**.
- G. BARONE and F. CINGANO, Service regulation and growth: evidence from OECD countries, Economic Journal, v. 121, 555, pp. 931-957, TD No. 675 (June 2008).
- P. SESTITO and E. VIVIANO, *Reservation wages: explaining some puzzling regional patterns*, Labour, v. 25, 1, pp. 63-88, **TD No. 696 (December 2008).**
- R. GIORDANO and P. TOMMASINO, *What determines debt intolerance? The role of political and monetary institutions*, European Journal of Political Economy, v. 27, 3, pp. 471-484, **TD No. 700 (January 2009).**
- P. ANGELINI, A. NOBILI e C. PICILLO, *The interbank market after August 2007: What has changed, and why?*, Journal of Money, Credit and Banking, v. 43, 5, pp. 923-958, **TD No. 731 (October 2009).**
- G. BARONE and S. MOCETTI, *Tax morale and public spending inefficiency*, International Tax and Public Finance, v. 18, 6, pp. 724-49, **TD No. 732 (November 2009).**
- L. FORNI, A. GERALI and M. PISANI, *The Macroeconomics of Fiscal Consolidation in a Monetary Union:* the Case of Italy, in Luigi Paganetto (ed.), Recovery after the crisis. Perspectives and policies, VDM Verlag Dr. Muller, **TD No. 747 (March 2010).**
- A. DI CESARE and G. GUAZZAROTTI, An analysis of the determinants of credit default swap changes before and during the subprime financial turmoil, in Barbara L. Campos and Janet P. Wilkins (eds.), The Financial Crisis: Issues in Business, Finance and Global Economics, New York, Nova Science Publishers, Inc., **TD No. 749 (March 2010).**

- A. LEVY and A. ZAGHINI, *The pricing of government guaranteed bank bonds*, Banks and Bank Systems, v. 6, 3, pp. 16-24, **TD No. 753 (March 2010).**
- G. BARONE, R. FELICI and M. PAGNINI, *Switching costs in local credit markets,* International Journal of Industrial Organization, v. 29, 6, pp. 694-704, **TD No. 760 (June 2010).**
- G. BARBIERI, C. ROSSETTI e P. SESTITO, The determinants of teacher mobility: evidence using Italian teachers' transfer applications, Economics of Education Review, v. 30, 6, pp. 1430-1444, TD No. 761 (marzo 2010).
- G. GRANDE and I. VISCO, A public guarantee of a minimum return to defined contribution pension scheme members, The Journal of Risk, v. 13, 3, pp. 3-43, **TD No. 762 (June 2010).**
- P. DEL GIOVANE, G. ERAMO and A. NOBILI, *Disentangling demand and supply in credit developments: a survey-based analysis for Italy*, Journal of Banking and Finance, v. 35, 10, pp. 2719-2732, **TD No.** 764 (June 2010).
- G. BARONE and S. MOCETTI, With a little help from abroad: the effect of low-skilled immigration on the female labour supply, Labour Economics, v. 18, 5, pp. 664-675, **TD No. 766 (July 2010).**
- S. FEDERICO and A. FELETTIGH, *Measuring the price elasticity of import demand in the destination markets of italian exports*, Economia e Politica Industriale, v. 38, 1, pp. 127-162, **TD No. 776 (October 2010).**
- S. MAGRI and R. PICO, *The rise of risk-based pricing of mortgage interest rates in Italy*, Journal of Banking and Finance, v. 35, 5, pp. 1277-1290, **TD No. 778 (October 2010).**
- M. TABOGA, Under/over-valuation of the stock market and cyclically adjusted earnings, International Finance, v. 14, 1, pp. 135-164, **TD No. 780 (December 2010).**
- S. NERI, *Housing, consumption and monetary policy: how different are the U.S. and the Euro area?*, Journal of Banking and Finance, v.35, 11, pp. 3019-3041, **TD No. 807** (April 2011).
- V. CUCINIELLO, *The welfare effect of foreign monetary conservatism with non-atomistic wage setters*, Journal of Money, Credit and Banking, v. 43, 8, pp. 1719-1734, **TD No. 810 (June 2011).**
- A. CALZA and A. ZAGHINI, welfare costs of inflation and the circulation of US currency abroad, The B.E. Journal of Macroeconomics, v. 11, 1, Art. 12, **TD No. 812 (June 2011).**
- I. FAIELLA, *La spesa energetica delle famiglie italiane*, Energia, v. 32, 4, pp. 40-46, **TD No. 822 (September 2011).**
- R. DE BONIS and A. SILVESTRINI, *The effects of financial and real wealth on consumption: new evidence from* OECD countries, Applied Financial Economics, v. 21, 5, pp. 409–425, **TD No. 837 (November 2011).**
- F. CAPRIOLI, P. RIZZA and P. TOMMASINO, *Optimal fiscal policy when agents fear government default*, Revue Economique, v. 62, 6, pp. 1031-1043, **TD No. 859** (March 2012).

2012

- F. CINGANO and A. ROSOLIA, *People I know: job search and social networks*, Journal of Labor Economics, v. 30, 2, pp. 291-332, **TD No. 600 (September 2006).**
- G. GOBBI and R. ZIZZA, Does the underground economy hold back financial deepening? Evidence from the italian credit market, Economia Marche, Review of Regional Studies, v. 31, 1, pp. 1-29, TD No. 646 (November 2006).
- S. MOCETTI, *Educational choices and the selection process before and after compulsory school*, Education Economics, v. 20, 2, pp. 189-209, **TD No. 691 (September 2008).**
- P. PINOTTI, M. BIANCHI and P. BUONANNO, *Do immigrants cause crime?*, Journal of the European Economic Association, v. 10, 6, pp. 1318–1347, **TD No. 698 (December 2008).**
- M. PERICOLI and M. TABOGA, *Bond risk premia, macroeconomic fundamentals and the exchange rate,* International Review of Economics and Finance, v. 22, 1, pp. 42-65, **TD No. 699 (January 2009).**
- F. LIPPI and A. NOBILI, *Oil and the macroeconomy: a quantitative structural analysis*, Journal of European Economic Association, v. 10, 5, pp. 1059-1083, **TD No. 704 (March 2009).**
- G. ASCARI and T. ROPELE, *Disinflation in a DSGE perspective: sacrifice ratio or welfare gain ratio?*, Journal of Economic Dynamics and Control, v. 36, 2, pp. 169-182, **TD No. 736 (January 2010).**
- S. FEDERICO, *Headquarter intensity and the choice between outsourcing versus integration at home or abroad*, Industrial and Corporate Chang, v. 21, 6, pp. 1337-1358, **TD No. 742 (February 2010).**
- I. BUONO and G. LALANNE, *The effect of the Uruguay Round on the intensive and extensive margins of trade*, Journal of International Economics, v. 86, 2, pp. 269-283, **TD No. 743 (February 2010).**

- S. GOMES, P. JACQUINOT and M. PISANI, The EAGLE. A model for policy analysis of macroeconomic interdependence in the euro area, Economic Modelling, v. 29, 5, pp. 1686-1714, TD No. 770 (July 2010).
- A. ACCETTURO and G. DE BLASIO, Policies for local development: an evaluation of Italy's "Patti Territoriali", Regional Science and Urban Economics, v. 42, 1-2, pp. 15-26, TD No. 789 (January 2006).
- F. BUSETTI and S. DI SANZO, *Bootstrap LR tests of stationarity, common trends and cointegration,* Journal of Statistical Computation and Simulation, v. 82, 9, pp. 1343-1355, **TD No. 799 (March 2006).**
- S. NERI and T. ROPELE, *Imperfect information, real-time data and monetary policy in the Euro area,* The Economic Journal, v. 122, 561, pp. 651-674, **TD No. 802 (March 2011).**
- G. CAPPELLETTI, G. GUAZZAROTTI and P. TOMMASINO, *What determines annuity demand at retirement?*, The Geneva Papers on Risk and Insurance – Issues and Practice, pp. 1-26, **TD No. 805 (April 2011).**
- A. ANZUINI and F. FORNARI, *Macroeconomic determinants of carry trade activity*, Review of International Economics, v. 20, 3, pp. 468-488, **TD No. 817 (September 2011).**
- M. AFFINITO, *Do interbank customer relationships exist? And how did they function in the crisis? Learning from Italy*, Journal of Banking and Finance, v. 36, 12, pp. 3163-3184, **TD No. 826 (October 2011).**
- R. CRISTADORO and D. MARCONI, *Household savings in China*, Journal of Chinese Economic and Business Studies, v. 10, 3, pp. 275-299, **TD No. 838 (November 2011).**
- P. GUERRIERI and F. VERGARA CAFFARELLI, Trade Openness and International Fragmentation of Production in the European Union: The New Divide?, Review of International Economics, v. 20, 3, pp. 535-551, TD No. 855 (February 2012).
- V. DI GIACINTO, G. MICUCCI and P. MONTANARO, Network effects of public transposrt infrastructure: evidence on Italian regions, Papers in Regional Science, v. 91, 3, pp. 515-541, TD No. 869 (July 2012).
- A. FILIPPIN and M. PACCAGNELLA, *Family background, self-confidence and economic outcomes,* Economics of Education Review, v. 31, 5, pp. 824-834, **TD No. 875 (July 2012).**

2013

- F. CINGANO and P. PINOTTI, *Politicians at work. The private returns and social costs of political connections*, Journal of the European Economic Association, v. 11, 2, pp. 433-465, **TD No. 709 (May 2009).**
- F. BUSETTI and J. MARCUCCI, *Comparing forecast accuracy: a Monte Carlo investigation*, International Journal of Forecasting, v. 29, 1, pp. 13-27, **TD No. 723 (September 2009).**
- D. DOTTORI, S. I-LING and F. ESTEVAN, *Reshaping the schooling system: The role of immigration*, Journal of Economic Theory, v. 148, 5, pp. 2124-2149, **TD No. 726 (October 2009).**
- A. FINICELLI, P. PAGANO and M. SBRACIA, *Ricardian Selection*, Journal of International Economics, v. 89, 1, pp. 96-109, **TD No. 728 (October 2009).**
- L. MONTEFORTE and G. MORETTI, *Real-time forecasts of inflation: the role of financial variables*, Journal of Forecasting, v. 32, 1, pp. 51-61, **TD No. 767 (July 2010).**
- E. GAIOTTI, Credit availablility and investment: lessons from the "Great Recession", European Economic Review, v. 59, pp. 212-227, TD No. 793 (February 2011).
- A. ACCETTURO e L. INFANTE, Skills or Culture? An analysis of the decision to work by immigrant women in Italy, IZA Journal of Migration, v. 2, 2, pp. 1-21, **TD No. 815 (July 2011).**
- A. DE SOCIO, *Squeezing liquidity in a "lemons market" or asking liquidity "on tap"*, Journal of Banking and Finance, v. 27, 5, pp. 1340-1358, **TD No. 819 (September 2011).**
- M. FRANCESE and R. MARZIA, is there Room for containing healthcare costs? An analysis of regional spending differentials in Italy, The European Journal of Health Economics (DOI 10.1007/s10198-013-0457-4), TD No. 828 (October 2011).
- G. BARONE and G. DE BLASIO, *Electoral rules and voter turnout*, International Review of Law and Economics, v. 36, 1, pp. 25-35, **TD No. 833 (November 2011).**
- E. GENNARI and G. MESSINA, How sticky are local expenditures in Italy? Assessing the relevance of the flypaper effect through municipal data, International Tax and Public Finance (DOI: 10.1007/s10797-013-9269-9), TD No. 844 (January 2012).
- A. ANZUINI, M. J. LOMBARDI and P. PAGANO, *The impact of monetary policy shocks on commodity prices*, International Journal of Central Banking, v. 9, 3, pp. 119-144, **TD No. 851 (February 2012).**

S. FEDERICO, *Industry dynamics and competition from low-wage countries: evidence on Italy*, Oxford Bulletin of Economics and Statistics (DOI: 10.1111/obes.12023), **TD No. 879 (September 2012).**

FORTHCOMING

- A. MERCATANTI, A likelihood-based analysis for relaxing the exclusion restriction in randomized experiments with imperfect compliance, Australian and New Zealand Journal of Statistics, TD No. 683 (August 2008).
- M. TABOGA, *The riskiness of corporate bonds*, Journal of Money, Credit and Banking, **TD No. 730** (October 2009).
- F. D'AMURI, *Gli effetti della legge 133/2008 sulle assenze per malattia nel settore pubblico*, Rivista di Politica Economica, **TD No. 787 (January 2011).**
- E. COCOZZA and P. PISELLI, Testing for east-west contagion in the European banking sector during the financial crisis, in R. Matoušek; D. Stavárek (eds.), Financial Integration in the European Union, Taylor & Francis, TD No. 790 (February 2011).
- R. BRONZINI and E. IACHINI, Are incentives for R&D effective? Evidence from a regression discontinuity approach, American Economic Journal : Economic Policy, **TD No. 791 (February 2011).**
- F. NUCCI and M. RIGGI, *Performance pay and changes in U.S. labor market dynamics*, Journal of Economic Dynamics and Control, **TD No. 800 (March 2011).**
- O. BLANCHARD and M. RIGGI, Why are the 2000s so different from the 1970s? A structural interpretation of changes in the macroeconomic effects of oil prices, Journal of the European Economic Association, **TD No. 835 (November 2011).**
- F. D'AMURI and G. PERI, Immigration, jobs and employment protection: evidence from Europe before and during the Great Recession, Journal of the European Economic Association, TD No. 886 (October 2012).
- R. DE BONIS and A. SILVESTRINI, *The Italian financial cycle: 1861-2011*, Cliometrica, **TD No. 936** (October 2013).